Notes of the RICH1 Phone meeting 06/07/2007
People Present:
CERN:  Fabio, Jon, Olav, Werner
IC:         Bill, Dave
Bristol:   Nick, Richard

Apologies for absence:
Neville, Rohan, Richard P
Discussed Agenda Topics:
1. Gas Enclosure

Richard P. has installed the temperature sensors in the G.E. The Milano group will install the aerogel boxes during the week starting the 16th of July, with Delio in attendance. Afterwards the GE needs to be cleaned prior to installation of the mirrors (vacuum cleaner and sticky tape roller). Once the mirrors are installed dry air will be used to keep them in a suitable clean atmosphere. Erich will connect a pipe from the dryer.  Temporary side doors do need to be put on after the mirror installation to protect the CF mirrors from accidental disturbance. The proper side doors, after removal of 4 bosses, will be put on for leak tests later on ~Sept07. 

2. Mirrors
Flat: The lower and upper flat mirrors have been aligned via survey to better than ~0.5 mrad.  The mirrors are now stored in their installation boxes in Bat. 156 until installation in RICH1. The V-blocks have been clamped to the rails using cable ties. The survey reports will be prepared by the survey team during the next weeks.
Spherical:  The spherical mirrors and the C-frames have been mounted onto the alignment rig and surveyed this week. Bill and Geoff come out to CERN to help. 4 mm have been machined off the bottom rail supports because the survey results indicated that the mirrors where 4mm too high.  The next steps are:
1) Monday afternoon: finish the optical bench alignment

2) Tuesday: align the optical system

3) Wednesday: align the mirrors to their CoCs

3) Friday:  survey the frame and mirrors, including the mirror gaps, after their alignment

The CMA mirrors will be monitored with cameras to check the D0 and the CoC spot position over a period of ~10 days. 
The horizontal gap between the mirrors is ~6mm while the vertical is ~3mm. The gap will be measured again after the alignment and if required could be reduced by using adjuster 1. 

Installation: 
The plan is to install the mirrors during week 23-27 July. Tim will come out to CERN to help. First the spherical mirrors will be installed because access into RICH1 is needed to check the mirror-beampipe gap. The CERN transport service has been already contacted and they will transport the mirrors to the LHCb pit (action Fabio).  A temporary storage area at the pit is required for the mirrors (action Dave, Fabio). After installation of the mirrors RICH1 will be filled with dry air.

Dave sent Bristol the drawings for the required modifications: the bosses on the RICH1 doors need to be trimmed due to the presence of the CMA label on the C-frame and the horizontal rig plates need to be drilled to serve as protection for the CF mirrors after installation. 
3. Aerogel
The Milano group will install the aerogel boxes during the week starting the 16th of July. Clara will coordinate things with Delio.
4. HPD
Column Dressing:   No news regarding the arrival of the cups. The last column has not yet been tested. The mechanics for the 2 spares will be in CERN in ~ 2 weeks.

Upper Rail System: NAB promised to deliver next week. A mistake was found which required re-welding. 
Testing/Mounting: The chariot manufacture at IC progresses. Some pieces are still missing. These should be completed in ~10 days. The chariot will be brought to Oxford for and transported to CERN with the rails. The system will be mounted and load tested probably around the beginning of August at point 8.

Lower Rail: Concept design has been sent by Brian to IC. Brian will be at IC next week to discuss things. Dave will be at CERN next week and will measure/check dimensions in the pit.
Cabling:  No news.
Light Leak detector:  No news.

Photon Funnel: No news.

5. HPD Cooling
There was an imbalance between the A and C sides of RICH2 which has now been corrected.  RICH1 will follow the RICH2 cooling scheme. Now both A and C sides of RICH2 are running at an input coolant temperature of 15 degrees
6. LAMS

P. Soler ordered all the required components. Expect to get the material in ~1 week. Plan to put the LAMS in RICH1 to monitor the spherical mirrors as soon as possible following installation of the mirrors.

7. MDMS

Have been in contact with Ray. Ray has prepared a plastic photon funnel mockup to test the mechanical stage. Several questions have been raised. The mechanics issues are nearly resolved now. The photon funnel will require some minor modifications. 
8. Quartz Windows
No news.
9. AOB: 

Werner reported that it is planned to test the magnet probably in September. The tests will be done in two steps. First the power converters need to be tested therefore the magnet needs to be switched on; the second step is the magnet commisioning.  September suits RICH1 but it would be preferable to have the two steps one after another without any long interval.
The last HPD was sent to Scotland yesterday. 

10. Next meeting
Friday 20th July 2007 at 10:00(GMT) 11:00(CET). 
